Author’s response to reviews

Title: Hazards of tube thoracostomy in patients on a ventilator

Authors:

Kasra Shaikhrezai (kasrash@gmail.com)
Vipin Zamvar (zamvarv@hotmail.com)

Version: 2 Date: 10 March 2011

Author’s response to reviews:

Dear Sir/Madam

As per your recent email regarding the major and minor revisions, we have revised the manuscript accordingly:

-Major revision: The phrase “all authors read and approved the final manuscript” has been added to the authors' contributions section.

Minor revision:

1- The words "case report" have been removed from the title page
2- The phrase “word count=” and the number have been removed from the title page
3- All references numbers are now in square brackets
4- In references section the journals names are printed in italic font with small changes of punctuation in favour of your journal formatting.